Dear Prof.

The manuscript “A Rare Finding of Primary Aortoduodenal Fistula by Gastrointestinal Bleeding SPECT/CT Scan: A Case Report” (WJCC_86582) has been revised according to the reviewer’s comments.

Looking forward to hearing from you soon.

Sincerely yours,

Yu-Hong Chang
Department of Nuclear Medicine,
Hsinchu Mackay Memorial Hospital,
Hsinchu 30013,
Taiwan, R.O.C.

Tel: 886-3-6119595
E-mail: 3053@mmh.org.tw
### Review Notes (WJCC_86582)

<table>
<thead>
<tr>
<th>Reviews’ comments</th>
<th>Revision/Change marked</th>
</tr>
</thead>
<tbody>
<tr>
<td><strong>Reviewer 1</strong></td>
<td><strong>Revision/Change marked</strong></td>
</tr>
<tr>
<td>This is an interesting case report, which is very rare, but it can still be found occasionally in patients with unexplained gastrointestinal bleeding. The manuscript demonstrates the importance of CTA and SPECT in diagnosis. However, the image quality of this manuscript is not very good and needs improvement.</td>
<td>I provided decomposable Figures (in which all components are movable and editable), organize them into a single PowerPoint file in accordance with WJCC regulation. The schematic diagram in Figure 4 has been reworked.</td>
</tr>
</tbody>
</table>
Dear authors,
It is very good manuscript about one of the rare cause of gastrointestinal bleeding. I read the manuscript attentively and have one comment: based on review of the literature it is better if you add investigation algorithm of this disease.

According to the WJCC template, I filled in the data step by step to provide a more complete investigation. A brief summary of diagnosis tools used for this disease is added to the conclusion section.